Author's response to reviews

Title: Influence of Gender, Sexual Orientation, and Need on Treatment Utilization for Substance Use and Mental Disorders: Findings from the California Quality of Life Survey

Authors:

Christine E. Grella (grella@ucla.edu)
Lisa Greenwell (lgreenwe@ucla.edu)
Vickie M. Mays (mays@ucla.edu)
Susan D. Cochran (cochran@ucla.edu)

Version: 4 Date: 3 June 2009

Author's response to reviews: see over
June 3, 2009

Sabina Alam, PhD
Scientific Editor
BMC Psychiatry

Re: MS no. 1177814652251787: Influence of Gender, Sexual Orientation, and Need on Treatment Utilization for Substance Use and Mental Disorders: Findings from the California Quality of Life Survey

Dear Dr. Alam,

We have made all of the formatting changes as requested in the email received on June 3, 2009 regarding the above manuscript. We look forward to its publication in BMC Psychiatry.

Sincerely,

Christine Grella

Christine E. Grella
Professor-in-Residence